Author's response to reviews

Title: Stores Healthy Options Project in Remote Indigenous Communities (SHOP@RIC): a protocol of a randomised trial promoting healthy food and beverage purchases through price discounts and in-store nutrition education.

Authors:

Julie K Brimblecombe (julie.brimblecombe@menzies.edu.au)
Megan M Ferguson (megan.ferguson@menzies.edu.au)
Selma C Liberato (selma.liberato@menzies.edu.au)
Kylie Ball (kylie.ball@deakin.edu.au)
Marjory L Moodie (marj.moodie@deakin.edu.au)
Anne Magnus (anne.magnus@deakin.edu.au)
Edward C Miles (emiles@icv.com.au)
Amanda J Leach (amanda.leach@menzies.edu.au)
Mark D Chatfield (mark.chatfield@menzies.edu.au)
Cliona Ni Mhurchu (c.nimhurchu@nihi.auckland.ac.nz)
Kerin O'Dea (Kerin.ODea@unisa.edu.au)
Ross S Bailie (ross.bailie@menzies.edu.au)

Version: 2 Date: 10 July 2013

Author's response to reviews:

Thank you for considering publication of the SHOP@RIC protocol.

Please find below my response to our queries:

Ethical approval and external funding:
- The study has ethical approval. Please find attached evidence of this from 3 HRECs in email.
- This study is funded by the Australian National Health and Medical Research Council (NHMRC; ID 1024285). (please see email with attachment)
- There are no other manuscripts under submission with other journals based on this study protocol.
- The trial is registered with a trial registry that is recognised by ICMJE. http://www.anzctr.org.au/. The trial registration number is: Australian New Zealand Clinical Trials Registry ACTRN12613000694718
- Title page: please find attached a copy of the full manuscript with title page included. Acknowledgements, authors contribution and funding details follow the discussion section of the manuscript. I have also uploaded the revised version to biomedcentral as instructed.

I trust this meets your approval,